Protected by copyright, including for uses related to text and data mining, Al training, and similar technologies.

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Understanding the Impacts of Care Farms on Health and Well-being of Disadvantaged Populations: A Protocol of the Evaluating Community Orders (ECO) Pilot Study
AUTHORS	Elsey, Helen; Bragg, Rachel; Elings, Marjolein; Cade, Janet; Brennan, Cathy; Farragher, Tracey; Tubeuf, Sandy; Gold, Rochelle; Shickle, Darren; Wickramasekera, Nyantara; Richardson, Zoe; Murray, Jenni

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.